Author's response to reviews

Title: Does Prophylactic Sotalol and Magnesium Decrease the Incidence of Atrial Fibrillation Following Coronary Artery Bypass Surgery: A Propensity-Matched Analysis

Authors:

Vikram Arrea (vikram.aerra@gmail.com)
Manoj Kudavalli (Manoj.Kudavalli@ctc.nhs.uk)
Adelard N Moloto (adelard2004@yahoo.fr)
Arun K Srinivasan (arunsri0@hotmail.com)
Anthony D Grayson (Tony.Grayson@ctc.nhs.uk)
Brian M Fabri (Brian.Fabri@ctc.nhs.uk)
Aung Y Oo (Aung.Oo@ctc.nhs.uk)

Version: 2 Date: 7 February 2006

Author's response to reviews: see over
Dear Vipin,

Thank you for your comments regarding our paper. I will answer your points in order.

1. Sotalol was administered orally and magnesium intravenously. We have included these corrections both in the abstract and in the text.
2. Cardiopulmonary bypass was included as one of the variables in the propensity score. This had been missed in the original submission and has now been corrected.
3. About 50% of all CABG cases are performed off bypass in the unit. Of course there is a significant variation between practices.
4. Spelling mistakes have been corrected.

The paper is now re-submitted with the appropriate corrections.